Author's response to reviews

Title: Cardiac incoordination induced by left bundle branch block: its relation with left ventricular systolic function in patients with and without cardiomyopathy

Authors:

Miguel Quintana (miguel.quintana@ki.se)
Samir Saha (samir.saha@ki.se)
Satish Govind (satish.govind@ki.se)
Francesca Del Furia (francesca_delfuria@hotmail.com)
Vicente Bertomeu (bertomeuv@gva.es)
Lars Å Brodin (lars-ake.brodin@syd.kth.se)

Version: 2 Date: 16 July 2008

Author's response to reviews: see over
Professor Eugenio Picano

Dear Editor:

On behalf of my co-authors, please receive enclosed a new version of the above mentioned manuscript.

The comments and issues raised by the referees have been taken into account when re-writing the manuscript as follows:

Referee 1:
  1. The results of the present manuscript are put in context with the results of the PROSPECT trial
  2. The conclusions in the abstract have been modified
  3. We do not have any own data supporting the value of myocardial contractile reserve as a predictor of response to CRT, however, the references suggested by the referee have been given in the present version
  4. Abbreviations in the manuscript have been replaced with full text
  5. We have tried to correct the typos
  6. A list of abbreviation is given

Referee 2:
  1. The typos have been corrected

Yours truly,

Miguel Quintana R. M.D., Ph.D.
The Karolinska Institute at the Department of Cardiology
Hospital de Torrevieja